Author's response to reviews

Title: The Calculation of Quality Indicators for Long Term Care Facilities in 8 countries (SHELTER project)

Authors:

Dinnus DHM Frijters (d.frijters@vumc.nl)
Henriette HG Roest van der (hg.vanderroest@vumc.nl)
Iain GI Carpenter (g.i.carpenter@kent.ac.uk)
Harriet H Finne-Soveri (harriet.finne-soveri@thl.fi)
Jean-Claude JC Henrard (jean-claude.henrard@sfr.fr)
Angela A Chetrit (zionch@bezeqint.net)
Jacob J Gindin (jacobgindin@hotmail.com)
Roberto R Bernabei (roberto_bernabei@rm.unicatt.it)

Version: 5 Date: 17 March 2013

Author's response to reviews:

Dear editor,

I received the following message from Nrs. Natalie Pafitis:

We note that you currently state within your manuscript that ethics approval for the study was obtained in all countries according to local regulations. In order for us to proceed, we would need for you to state specifically within your manuscript that ethics approval for the study was obtained in all countries and specifically from the following ethics committees (METC VUmc, METC Universita Cattolica del Sacro Cuore, etc *list remaining ethics committees*)

I have now changed the submitted paper accordingly. I hope that you now will be able to proceed with its publication.

Best regards

Dinnus Frijters